Author's response to reviews

Title: Idiopathic combined, autoantibody-mediated ADAMTS13-factor H deficiency in TTP-HUS

Authors:

Daniel Patschan (d.patschan@gmail.com)
Peter Korsten (peter.korsten@web.de)
Arne Behlau (Arnea.behlau@web.de)
Radovan Vasko (rvasko@inbox.com)
Malte Heeg (malte.heeg@med.uni-goettingen.de)
Nadera Sweiss (nsweiss@medicine.bsd.uchicago.edu)
Gerhard A Müller (nephrorheuma@med.uni-goettingen.de)
Michael Koziolek (mkoziolek@med.uni-goettingen.de)

Version: 3 Date: 10 October 2011

Author's response to reviews: see over
Dear Prof. Kidd,

Herewith we would like to submit a revised version of our case report with the title:

**Idiopathic combined, autoantibody-mediated ADAMTS13/factor H deficiency in TTP-HUS**

We were able to address all concerns and suggestions expressed in the reviews. Specifically, the following modifications were implemented:

1. the discussion sections has significantly been shortened
2. information about the patient’s ethnicity has been added
3. information about the written informed consent by the patient’s legal guardian has been added.

We hope that the revised version of the manuscript will be acceptable for the publication in the Journal.

Sincerely,

D. Patschan, MD PhD